 Vc/17-May-04
Week21 starting 17 May 04

SM18 Operation Meeting (Tuesdays)

 Bench status  and tests for now & next few days : 

	Bench
	Name
	Situation Mon 17/5 afternoon
	Temp

	TBA1
	MBAR3061
	Prep 2.2 IAP @ warm LF
	1.9 K

	TBA2
	MBBL3117
	PT 1 IAP @ cold
	4.4 K

	TBB1
	MBBL1104
	ICS 2 Connect Magnet (ICS)
	300 K

	TBB2
	MBAL1071
	PT 12 Warm Up
	Warming

	TBC1
	MBBL1116
	New Magnet arrived
	300 K

	TBC2
	SSS21
	PT 5.2 1.5 KA Quench HF
	1.9 K

	TBD1
	MBBL2076
	PT 6.5 Training 5
	1.9 K

	TBD2
	MBBL2062
	ICS 1 WP04 HV Test Warm
	300 K

	TBE1
	MBBL1112
	ICS 4 Final connection
	300 K

	TBE2
	MBBL1113
	PT 6.7 Training 7
	1.9 K

	TBF1
	MBBL3151
	ICS 1 WP04 HV Test Warm
	300 K

	TBF2
	MBAL3160
	PT 6.1 Training 1
	 


Test Coordinator this week: Pierre ; MTM support on call  Federico:  (16-3602).

7 Dipoles  Disconnected after tests last week    ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· D2 Cryo Repair Last week of faulty Valve
A2  commissioning   & re-test     3117 
A1         “                   &   re-test    3061
· D1 & D2  issues  ( Jurek) 
· SSS work :        To Do list on Board+ GREY FILE IN SMCR  
· General Problems last week :  
	Date
	Bench/

magnet
	Problem
	Cause
	Solution

	Fri 14th aft
	B1  (2039)
	Power supply pblm for T-coil
	Poor Config.
	Giorgio fixed it

	Fri 
14th May

night
	D2 (2078)
	Training quench at 7969A & down to ZERO without any reason
	TEMA error like before ???
	quench ignored, training continued normally

	Sat 
15th May
	C1 (3504)
	Last HV cold test incomplete, discovered when magnet was already warming up (~40 K)
	Human error – check with Khare? 
	Magnet cooled back down to 4.2, HV test completed

	Sat 15th night
	D Cluster
	No label tags on Dipole leads at CFB level
	
	Giorgio informed Mon.17th May

	Sun 
16th May
	E2 (1113)
	AQA timing faults on 1.5kA quenches
	Slow quench development, most of signal lost due to insufficient 3s pre-trigger length
	Quench repeated, acquisition worked successfully

	Sun 
16th May
	E2 (1113)
	Training quench at ~11kA
	Unexplained V-clamp quench
Cryo OK signal active
	Quench ignored, training continued normally

	Sat-Sun
	All
	TV status display crashed
	Network/PC problems on Jerome’s DB ?
	Pending

	Sat-Sun
	All
	Water interlock on D-E cluster, no more than 2 Converters active (Flo)
	Human error ?
	Richard switched on all valves in the correct order








